communications medicine

ARTICLE

https://doi.org/10.1038/s43856-022-00124-5

OPEN

Antibiotic-resistant organisms establish reservoirs in new hospital built environments and are related to patient blood infection isolates

Kimberley V. Sukhum^{1,2,7}, Erin P. Newcomer ^{1,2,3,7}, Candice Cass⁴, Meghan A. Wallace², Caitlin Johnson², Jeremy Fine², Steven Sax⁴, Margaret H. Barlet ⁴, Carey-Ann D. Burnham^{2,4,5,6 K}, Gautam Dantas ^{1,2,3,5 K} & Jennie H. Kwon ^{4 K}

Abstract

Background Healthcare-associated infections due to antibiotic-resistant organisms pose an acute and rising threat to critically ill and immunocompromised patients. To evaluate reservoirs of antibiotic-resistant organisms as a source of transmission to patients, we interrogated isolates from environmental surfaces, patient feces, and patient blood infections from an established and a newly built intensive care unit.

Methods We used selective culture to recover 829 antibiotic-resistant organisms from 1594 environmental and 72 patient fecal samples, in addition to 81 isolates from blood cultures. We conducted antibiotic susceptibility testing and short- and long-read whole genome sequencing on recovered isolates.

Results Antibiotic-resistant organism burden is highest in sink drains compared to other surfaces. *Pseudomonas aeruginosa* is the most frequently cultured organism from surfaces in both intensive care units. From whole genome sequencing, different lineages of *P. aeruginosa* dominate in each unit; one *P. aeruginosa* lineage of ST1894 is found in multiple sink drains in the new intensive care unit and 3.7% of blood isolates analyzed, suggesting movement of this clone between the environment and patients.

Conclusions These results highlight antibiotic-resistant organism reservoirs in hospital built environments as an important target for infection prevention in hospitalized patients.

Plain Language Summary

Check for updates

Patients in hospitals often have a suppressed immune system, putting them at increased risk of infection by bacteria that are resistant to antibiotics, some of which may come from sources in the hospital environment. We sampled multiple different surfaces in an established and a newly built intensive care unit and collected patient infection samples. We tested bacteria in these samples for their resistance to antibiotics and sequenced the genetic code of the bacteria to identify relationships between environmental and patient infections. We found the most antibiotic resistant organisms in hospital sink drains. Our sequencing data revealed strains of a certain kind of bacteria could form reservoirs and survive in sink drains and also cause patient infections. These results highlight the importance of removing these antibiotic resistant organism reservoirs to prevent infections.

¹ The Edison Family Center for Genome Sciences and Systems Biology, Washington University School of Medicine in St Louis, St Louis, MO, USA. ² Department of Pathology and Immunology, Washington University School of Medicine in St Louis, St Louis, MO, USA. ³ Department of Biomedical Engineering, Washington University in St Louis, St Louis, MO, USA. ⁴ Department of Medicine, Washington University School of Medicine in St Louis, St Louis, MO, USA. ⁵ Department of Molecular Microbiology, Washington University School of Medicine in St Louis, St Louis, St Louis, MO, USA. ⁶ Department of Pediatrics, Washington University School of Medicine in St Louis, St Louis, St Louis, St Louis, MO, USA. ⁷ These authors contributed equally: Kimberley V. Sukhum, Erin P. Newcomer. ^{Ke}email: cburnham@wustl.edu;dantas@wustl.edu;j.kwon@wustl.edu

ealthcare-associated infections (HAIs) are a global challenge, posing a particularly acute threat in intensive care units (ICUs) where critically ill and immunocompromised patients are at elevated risk for infection during their stay^{1,2}. Worldwide, HAIs are responsible for an estimated 2.5 million infections every year and are associated with increased morbidity, mortality, and healthcare costs^{1,3–5}. The COVID-19 pandemic is associated with further expansion of hospitalized critically-ill individuals⁶. HAIs due to AROs in the ICU can be difficult to treat due to limited treatment options; available options are also associated with toxicity, are poorly tolerated by patients, and may exhibit negative interactions with other drugs^{1,7,8}.

Many studies and initiatives have focused on trying to limit HAIs through surveillance, prevention, and intervention^{1,9,10}. Recent studies have used culture-independent metagenomic sequencing of hospital surfaces to generate an important catalog of the diversity and composition of their resident microbial communities¹¹⁻¹⁵. However, metagenomic characterizations are limited in their ability to track viable, antibiotic-resistant strains and remain ambiguous to whether the taxa discovered on surfaces are environmental- or patient-derived, and/or associated with infections in patients. To better understand relationships between viable antibiotic-resistant organisms (ARO) in the built environment and critically-ill patients, we must determine 1) what hospital surfaces are acting as ARO reservoirs, i.e., surfaces where an organism can be cultured from multiple time points; 2) what are the spatial and temporal dynamics of reservoir colonization; and 3) whether viable ARO strains colonizing the hospital built environment can also be detected from human clinical infections.

There are multiple models proposed for ARO reservoir colonization and transmission in hospitals (Fig. 1a)^{1,16-18}. A prominent model is that AROs are shed from colonized patients, frequently through fecal contamination, to surfaces, instruments, and shared equipment in patient rooms (Fig. 1a)^{19,20}. High-touch hospital surfaces can act as intermediate ARO reservoirs, and transmission may occur from these reservoirs through patients, healthcare staff, and visitors^{10,20-23}. Another model is that AROs are seeded from microbial communities which persistently colonize hospital built environments, particularly plumbing sources, where biofilms form and can act as a reservoir for potential pathogens (Fig. 1a)²⁴⁻²⁷. These models are not mutually exclusive. ARO reservoirs are likely dependent on a given facility's history and modes of transmission likely interact within a hospital²⁸. To better understand the colonization and transmission of AROs in the hospital built environment, we leveraged a unique opportunity to sample a newly-built stem cell transplant and oncology (SCT) ICU both before patient and staff occupancy and for one year after ICU establishment. This allowed us to identify and track persistent colonization of sink drains by AROs that began prior to patient and staff occupancy, a facet that has not been characterized in previous studies. As immunocompromised cancer patients demonstrate prolonged duration of ARO shedding and are at high risk of HAIs, the SCT ICU is a critical environment to study ARO surface colonization and transmission²⁹⁻³². Additionally, we compared this new ICU environment (new ICU) with environmental samples from the established SCT ICU previously housing these patients and staff (old ICU). While previous studies have longitudinally tracked surface and patient samples within an ICU, they have been limited in their ability to discern the impact of the facility built environment from the population of patients and healthcare workers in the facility. Here, the same patients and healthcare providers transitioned between the old and new buildings across the study period, allowing for a direct comparison between their ARO communities.

To track ARO transmission events between patients and ICU surfaces, we collected remnant fecal samples from patients in the SCT ICU who had laboratory studies ordered on fecal samples and isolates from positive blood cultures ordered as part of routine clinical care during the same collection period. From this unique collection of environmental and patient samples, we used selective microbiologic culturing and whole-genome sequencing (WGS) to identify AROs, assess antibiotic resistance, and track strains across time and location.

We found ARO contaminants were rare on most ICU surfaces but prevalent in sink drains in both ICUs, with the old ICU having significantly higher ARO burden in sink drains than the new ICU. AR Enterobacterales, which are frequently associated with fecal contamination, were rarely found on surfaces. In both ICUs, Stenotrophomonas spp. and Pseudomonas spp. were the two most frequently collected genera; however, different lineages dominated each ICU. Stenotrophomonas maltophilia strains formed months-long reservoirs in sink drains in the new ICU with no evidence of strains association with bloodstream infections during our study time period. In contrast, Pseudomonas aeruginosa strains formed persistent reservoirs for most of the year in the new ICU in multiple sink drains and showed evidence of shared strains across environmental samples and patient blood cultures. These results provide evidence that sink drains in the healthcare environment can serve as ARO reservoirs that are associated with human clinical infections.

Methods

Sample collections and culturing. Environmental and fecal samples received a non-human subjects determination by the Institutional Review Board (IRB) of Washington University (201712083). Blood culture clinical isolate collection was reviewed and approved by IRB (201901053) and by the Siteman Cancer Center Protocol Review and Monitoring Committee. We received IRB approval and Siteman Cancer approval for clinical isolates from patients. The IRB granted a waiver of informed consent for the collection of these specimens because they had been collected as part of routine clinical care. We sampled 6 SCT ICU (old ICU) rooms 3 times over the course of 1 month in the old building from December 2017 - January 2018. At each time point, nine surfaces were sampled using Eswab collections (Copan) pre-moistened with molecular water: the foam dispenser, the gown and glove storage area, the bedside rail, the nursing call button, the room floor, the light switch, the computer, the inroom sink handles, and the in-room sink drain. Three swabs were held together to simultaneously sample each surface. We also collected 2 samples of 15 mL in-room sink water directly from the faucet: 1 sample was collected immediately after turning the faucet on, and 1 sample was collected after allowing the water to run for 2 min.

We sampled 6 SCT ICU (new ICU) rooms and communal SCT ICU areas every other week for 5 months and then every month for 1 year in the new building for a total of 21 samplings (Fig. 1b). Samples were collected twice during the first week of sample collections in the new ICU building: the first after construction terminal clean and the second after custodial terminal clean. Both time points collected were before patients and staff had entered the ICU. At each time point, the same nine patient room surfaces as described above were sampled plus an additional 3 surfaces: the sofa from the patient room, the bathroom toilet from adjoining bathroom, and the sink drain from the adjoining bathroom. We also collected 15 mL of in-room sink water and bathroom sink water. At each time point, we also sampled four communal surfaces: the housekeeping closet drain, the family area floor, the soiled utility room drain, and the vending machine. For each time



Fig. 1 ARO reservoir colonization models and sample processing scheme. a Two models of reservoir colonization. Model 1 shows antibiotic-resistant organism (ARO) transmission from patients to hospital surfaces and then to other patients. Model 2 shows ARO transmission from environmental reservoirs to hospital surfaces to patients. **b** Sample collection time points and sample processing scheme from surface collections to WGS. In sample collection scheme, large circles represent months with small circles representing 2-week sampling within months. Purple indicates old intensive care unit (ICU) collections, green indicates new ICU collections, and pink indicates collections taken before patients enter the building in the new ICU. Icons labeled as such were acquired from nounproject.com, and other icons were used with permission from D'Souza, Potter et al.⁸². AST antibiotic susceptibility testing, MALDI-TOF matrix-assisted laser desorption/ionization-time of flight mass spectrometry.

point in both buildings, we obtained remnant de-identified fecal specimens that had been submitted to the clinical microbiology laboratory for *C. difficile* testing from patients in the same unit as surface swab collection.

Eswab specimens from surfaces, water samples and fecal samples were cultured the same day of sampling. Eswab

specimens were vortexed and 90 μ L of eluate was used for culture inoculation per plate/test condition. For fecal specimens, 90 μ L of specimen was used for culture inoculation. For water samplings, 100 μ L of vortexed water sample was used for culturing. All samples were inoculated to each of the following culture medium: Sheep's blood agar (Hardy), VRE chromID (bioMerieux), Spectra

MRSA (Remel), HardyCHROM ESBL (Hardy), MacConkey agar with cefotaxime (Hardy), Cetrimide agar (Hardy), and Sabouraud dextrose + chloramphenicol (Hardy). Plates were incubated at 35 °C in an air incubator and incubated up to 48 h prior to discard if no growth (up to 7 days for sabouraud dextrose + chloramphenicol). Two colonies of each colony morphotype were subcultured and identified using matrix-assisted laser desorption/ ionization time-of-flight mass spectrometry (MALD-TOF MS) with the VITEK MS system. All isolates recovered were stored at -80 °C in TSB with glycerol.

Isolates recovered from standard-of-care blood cultures during the same time frame of the surface sampling were recovered from frozen stocks in the clinical microbiology laboratory.

Antimicrobial susceptibility testing. Antimicrobial susceptibility testing (AST) was performed using Kirby Bauer disk diffusion, interpreted according to CLSI standards³³. AST was performed on gram negative bacilli using ampicillin, cefazolin, cefotetan, ceftazidime, ceftriaxone, cefepime, meropenem, ciprofloxacin, levofloxacin, piperacillin-tazobactam, ceftolozane-tazobactam, ceftazidime-avibactam, ampicillin-sulbactam, trimethoprim-sulfamethoxazole, gentamicin, amikacin, fosfomycin, colistin, aztreonam, doxycycline, minocycline, and nitrofurantoin and antimicrobials were interpreted/reported as appropriate for the specific species. We also performed a carbapenamase inactivation assay on all Enterobacterales and *Pseudomonas* isolates that were resistant or intermediate to meropenem or imipenem.

Short read sequencing. Total genomic DNA was extracted from cultured isolates using the Bacteremia kit (Qiagen, Gernmantown, MD, USA) and DNA was quantified using the PicoGreen dsDNA assay (Thermo Fisher Scientific, Waltham, MA, USA). A total of 5 ng/µL was used as input for Illumina sequencing libraries with the Nextera kit (Illumina, San Diego, CA, USA). The libraries were pooled and sequenced on a NextSeq HighOutput platform (Illumina) to obtain 2x150bp reads. The reads were demultiplexed by barcode and had adapters removed with Trimmomatic³⁴. Reads are available under BioProject PRJNA741123 (http://www.ncbi. nlm.nih.gov/bioproject/741123). Processed reads were assembled into draft genomes using SPAdes v3.11.035. Assemblies were assessed for quality using Quast v3.236 and checkM v1.0.1337. Assemblies were considered to have passed quality standards if completeness was greater than 90% and contamination was below 5%. We used Prokka on the assembled genomes to identify and annotate open reading frames³⁸.

Long read sequencing. Isolates were streaked from frozen stocks onto LB agar and allowed to grow at 37 °C for 48 h prior to extraction. Lawns were scraped from plates into nuclease free water. Genomic DNA was extracted using the bacteremia kit (Qiagen, Gernmantown, MD, USA), with the modification of limiting the vortex step to 2 min to preserve DNA fragment length. A total of 1 ug DNA from each isolate was used as input for library preparation using the Oxford Nanopore ligation sequencing kit and native barcode expansion kits (Oxford Nanopore Technologies, Oxford Science Park, OX4 4DQ, UK). Libraries were pooled and sequenced on a MinION flow cell (Oxford Nanopore Technologies, Oxford Science Park, OX4 4DQ, UK). Raw reads were preprocessed using Filtlong v0.2.039 with parameters -min_length 1000 -keep-percent 95 -target_bases 650000000. Hybrid assemblies were created by assembling long read sequencing data in Flye v2.8.140 and polished with short reads from Illumina sequencing⁴¹. Assemblies were assessed for quality using Quast v3.2³⁶ and checkM v1.0.13³⁷. Reads are available under BioProject PRJNA741123 (http://www.ncbi.nlm. nih.gov/bioproject/741123).

Genomic taxonomic identification. Following draft assembly, we determined taxonomic identification by ANI, MASH, and MLST. Species were determined if the genome had >75% aligned bases and >95% ANI with the type genome. Assembled genomes were considered to be the same genomospecies if they had >95% pairwise match but no >95% match with a type genome. We compared all assembled genomes against all assembled genomes and all type genomes using dnadiff⁴². If no species were identified, we used Mash to determine genera by comparing assembled genomes against all NCBI reference genomes⁴³. After all phages were removed, genera were considered to be the same as the hit/ hits with the highest identity. MLST was determined using mlst v2.4^{44,45}.

Phylogenetic analyses. To create core genome alignments, the gff files produced by Prokka were used as input in Roary⁴⁶. Roary alignments were used to create an approximate maximum like-lihood tree with FastTree⁴⁷. Branch length precision was rounded to 0.0001 substitutions per site. The output newick files were visualized and annotated with isolate source data using ggtree (R)^{48,49}. Roary pangenome sequences were further annotated using EggNOG v5.0⁵⁰.

Isolate groupings based on SNP pairwise distances. Snippy v4.4.3⁵¹ was used to map forward and reverse reads for isolates to the type strain complete genome assembly and to call SNPs. To determine groups, we compared pairwise SNP distances between each isolate pairs of the same species. Isolates were grouped into perfectly reciprocal groups at every pairwise distance cutoff between isolates using igraph⁵². The SNP distance cutoff was set at the lowest SNP value where number of groups plateaued for many thousands of SNPs, indicating that the members of these groups are much more closely related to one another than other isolates.

Antibiotic-resistant gene identification and analyses. We identified acquired antibiotic resistance mutations against aminoglycosides, amphenicols, β -lactam, folate pathway inhibitors, fosfomycin, macrolides/lincosamides/streptogramins, quinolones, rifamycin, tetracycline, and vancomycin using ResFinder⁵³.

Bayesian phylogenetic analysis of molecular sequences using BEAST 2. Group 1 isolates were long-read sequenced and quality filtered as described above, and the core genome alignment was constructed as above. The core genome alignment was composed of 5964 core genes out of 6986 total genes, which we used as input genes for our time-measured phylogenetic analysis in BEAST v2.6.5⁵⁴. The core genome alignment was converted to a Nexus file using MEGA X⁵⁵. We used BEAUti v2.6.5 from the BEAST v2.6.5⁵⁴ software package to convert the Nexus file into a.xml file for input into BEAST. We chose to use the HKY site model because it allows for some flexibility in substitution rate for different types of substitutions, and catches most major biases⁵⁶. We also used a strict clock model because our sequences are all from the same hospital within just over a year of each other, so we have no reason to suspect different substitution rates for different lineages⁵⁶. Tip dates were determined as the number of days between each sample and the first sample collected. Model diagnostic information and parameter distribution were viewed using Tracer v1.7.2⁵⁷. Individual trees were visualized using FigTree v1.4.4⁵⁸ and the consensus tree was visualized using DensiTree v2.2.7⁵⁹.



Fig. 2 Variation in isolate collection location, identity, and timing across all sampling. Error bars indicate standard error of intensive care unit (ICU) rooms. ** indicates generalized linear mixed-modeling (GLMM) *p*-value <0.01. **a** In-room and bathroom sink drains have significantly more isolates per collection than other surface locations in both the old and new ICU buildings (n = 566 surface isolates). Locations in light gray were not collected in old ICU. **b** Genus of matrix-assisted laser desorption/ionization-time of flight mass spectrometry (MALDI-TOF) species identification of all collected isolates in both the new and old ICU. Other Surface includes all other surfaces that are not in-room or bathroom sink drain. **c** Variation in number of isolates collected per bathroom or in-room sink drain sample collection by building (excludes fecal and communal samples, n = 429). **d** Variation in number of isolates per other surface sample collection for all time points, n = 429. **f** Variation in number of isolates callected across all time points (excludes sink drain, fecal, and communal samples, n = 137). **e** variation in number of surfaces. BP before patient and staff move-in.

Statistics and reproducibility. Comparative statistics between old and new building samples were normalized by number of samplings. Generalized linear mixed models were used for significance testing, with Room and Week as random effects. In Fig. 2c, d, isolate frequencies were collapsed by Room and then averaged. Error bars indicate standard error. For all main text phylogenetic trees, branches with less than 80% bootstrap support were collapsed, and branches with 80–90% bootstrap support were labeled as such. Supplementary Figures containing phylogenetic trees (Supplementary Figs. 1c, d, 2, and 3a) have a minimum resolution of 0.00055.

Reporting summary. Further information on research design is available in the Nature Research Reporting Summary linked to this article.

Results

AROs were collected and cultured from ICU surfaces, fecal samples, and clinical blood cultures in an old and new ICU. To test models of ARO reservoir colonization and transmission in a hospital built environment setting, we collected 1594 surface samples and 72 patient fecal samples at 24 time points from 6 ICU rooms in 2 buildings. Full metadata for 829 collected isolates has been included as Supplementary Data 1. The first building was the SCT ICU that was located in a well-established hospital building, the old ICU. The second was a newly constructed SCT ICU (new ICU); after construction was completed on the new ICU, the same staff and patients from the old ICU were all relocated to the new ICU. The old ICU rooms were sampled 3 times, with a week between samplings, during the final month of ICU occupancy (Fig. 1b). New ICU rooms were sampled twice (two days apart) after the completion of construction while the

rooms were unoccupied, then once every other week for the first 5 months of patient and staff occupancy (n = 11 samplings), then once every month for the rest of the first year of occupancy (n = 8 samplings) (Fig. 1b). For both ICUs, we swabbed 10 hightouch ICU surfaces (with an additional 4 surfaces from attached bathrooms in new ICU rooms). We also obtained remnant fecal samples submitted for routine *Clostridioides difficile* testing as well as isolates recovered from standard-of-care blood cultures from patients in the ICU. We utilized selective microbiologic culture on surface and fecal samples to enrich for and culture AROs, including 1) organisms that form colonies on antibiotic media, which we later assessed for resistance phenotypes by antibiotic susceptibility testing (AST), and 2) organisms that are inherently resistant to antibiotics, including Pseudomonas, Stenotrophomonas, and C. difficile⁶⁰⁻⁶². Results from AST can be found in Supplementary Data 2. Blood culture isolates were recovered in the clinical laboratory as part of routine clinical methods (i.e., not selectively cultured for ARO) and were retrospectively obtained for during 46 different weeks of the study spanning 61 weeks total. We recovered 566 AROs from surface environmental samples and 164 AROs from fecal samples, and we obtained 99 isolates from clinical blood cultures in the clinical microbiology laboratory.

Sink drains had a high ARO burden compared to other ICU surfaces. To identify potential ARO surface reservoirs, we measured ARO burden (defined as number of different ARO isolates/ morphotypes per samples collected) on different surfaces. Cultured bacteria were identified using VITEK MS matrix assisted laser desorption ionization-time of flight mass spectrometry (MALDI-TOF MS) (bioMerieux). ARO burden was significantly higher in sink drains than on other ICU room surfaces in both

the old and new ICUs (Fig. 2a, GLMM: p < 0.001, marginal $R^2 = 0.942$, conditional $R^2 = 0.945$). All other ICU room surfaces had at least a 6-fold lower ARO burden (mean old ICU sink drains: 4.02 isolates/sample collection, mean old ICU other surfaces: 0–0.64 isolates/sample collection, mean new ICU sink drains: 1.59–1.72, mean new ICU other surfaces: 0–0.21). ICU sink water had low ARO burden (mean range of 0–0.02 isolates/ sample collection) (Fig. 2a).

Pseudomonas was the most frequently detected genus, comprising 235/696 (33.8%) of all isolates cultured from the new ICU and 48/133 (36.1%) from the old ICU (Fig. 2b). The second most frequently identified genus was Stenotrophomonas (115/696 (16.5%) in the new ICU and 13/133 (9.8%) in the old ICU). Both genera were found primarily in sink drain samples (215/283 (76.0%) of Pseudomonas and 114/128 (89.1%) of Stenotrophomonas). Enterobacterales made up 77/696 (11.1%) and 20/133 (15.7%) of all isolates from the new and old ICUs, respectively, but only 7/97 (7.2%) were isolated from surface samples (Fig. 2b). Candida spp. isolates were identified in both the new ICU (64/696 (9.2%)) and old ICU (12/133 (9.0%)) with isolates primarily coming from fecal samples (50/76 (65.8%)) and room floor (10/76 (13.2%)). Gram positive AROs, including genera Enterococcus, Staphylococcus, and Clostridium, were found in both the new ICU (63/696 (9.0%)) and old ICU (23/133 (17.3%)). Clostridium was recovered from in fecal samples (n = 5). Staphylococcus and Enterococcus were found primarily in blood and fecal samples (52/81 (64.2%) of Staphylococcus and Enterococcus) and never found in sink drains. (Fig. 2b). These data suggest that in both buildings, sink drains are areas of substantial concern since they persistently yield cultures of Pseudomonas spp. and Stenotrophomonas spp., which both include strains capable of causing human infection^{63,64}.

ARO burden did not increase after patients and staff move in or over one year of sampling in new ICU. Next, we compared ARO burden across ICUs, patient and staff occupancy, and time points. Since there were large differences in ARO burden across surfaces, we separated environmental samples into 2 groups: sink drains and other (Fig. 2c, d). First, we compared ARO differences between the old ICU and new ICU before and after patient occupancy. We found ARO burden was higher in the old ICU than in the new ICU in sink drains (Fig. 2c, GLMM p < 0.001, $R^2 = 0.59$) but not on other surfaces (Fig. 2d, GLMM p > 0.05, $R^2 = 0.07$). Further, there was no difference in ARO burden before and after patient occupancy (Fig. 2c, d, GLMM p > 0.05). When we compared ARO burden in sink drains over time, we found no significant differences between the first week of collection after patients' occupancy in the new ICU and any other time point collected (Fig. 2e, Wilcoxon signed-rank test p > 0.05). The same was true for other surface collections (Fig. 2f, Wilcoxon signed-rank test p > 0.05), although ARO burden for other surfaces had high variation across weeks (mean range 0.02-0.35). Together, this suggests that there were environmental-associated differences in ARO burden between the old and new ICUs, and that ARO burden did not change after patient occupancy in the new ICU nor significantly increase or decrease during 1 year of collections.

No evidence of AR Enterobacterales reservoirs on surfaces in either ICU. To determine taxa-specific patterns in reservoir colonization, we performed WGS of Enterobacterales, *Pseudomonas*, and *Stenotrophomonas* isolates from environmental, fecal, and blood samples from both ICUs. AR Enterobacterales are some of the most feared AROs for HAIs⁸ and many are associated with human fecal colonization $^{65-67}$. We collected 97 isolates from 4 genera of Enterobacterales: Escherichia, Klebsiella, Citrobacter, and Enterobacter (Supplementary Fig. 1A). Isolates were recovered primarily from fecal samples (45/97 (46.4%) of Enterobacterales) and from blood cultures (45/97 (46.4%) of Enterobacterales) (Supplementary Fig. 1A). Escherichia coli was the most frequently detected Enterobacterales species (37/97 (38.1%)), followed by Klebsiella pneumoniae (18/97 (18.6%)) (Supplementary Fig. 1A). Notably, from 1594 surface samples over 24 time points, there were only 7 instances of an Enterobacterales isolate being cultured from an ICU surface sample (Supplementary Fig. 1B). Of the 7 isolates, 2 were different morphotypes of Citrobacter freundii isolated from the same sample with high average nucleotide identity (ANI) (99.99%), suggesting closely-related organisms or morphovariants. Apart from those 2 C. freundii isolates, no 2 surface Enterobacterales were the same species and no 2 Enterobacterales were found on the same surface twice (Supplementary Fig. 1B). These data suggest AR Enterobacterales do not represent ARO reservoirs on any of the sampled ICU surfaces, despite being present in many patient fecal samples.

To determine within species isolate similarity, we compared strain genomes and antibiotic resistance profiles across the two most frequent Enterobacterales species: E. coli and K. pneumoniae. When we compared multi-locus sequence typing (MLST) profiles of E. coli isolates, we found one instance of shared sequence type (ST131) between a surface isolate and a blood or fecal isolate. In a core genome phylogenetic tree, we found no phylogenetic clustering based on isolate type or ICU, except for 3 different *E. coli* morphotype isolates all taken from the same fecal sample and sharing 99.98% ANI (Supplementary Fig. 1C). To determine if antibiotic resistance profiles vary by sample type or location, we determined phenotypic susceptibility and identified antibiotic resistance genes (ARGs) using Resfinder^{68,69}. By Kirby Bauer disk diffusion, interpreted according to Clinical and Laboratory Standards Institute (CLSI) standards, 2/37 blood E. coli isolates were not resistant or intermediate resistant to any tested antibiotics. AST profiles varied across the E. coli isolates with isolates frequently resistant to ampicillin (23/37), cefazolin (20/37), ciprofloxacin (19/37), and levofloxacin (19/37) (Supplementary Fig. 1C). We found 9/37 E. coli isolates were resistant to cefepime, including the 1 surface isolate, and no E. coli isolates were resistant to meropenem. We found 24/37 E. coli isolates were resistant to multiple antibiotics with 20 isolates resistant to four or more drugs. E. coli isolates harbored various ARGs (Supplementary Fig. 1C), but ARG profile did not vary by sample type or location.

In K. pneumoniae isolates, we also found no phylogenetic clustering based on isolate type or ICU building (Supplementary Fig. 1D). Only one K. pneumoniae isolate, which was recovered from patient blood culture, demonstrated meropenem resistance, but it was negative for carbapenemase activity using the Carbapenem Inactivation Method³³. 3/18 K. pneumoniae isolates were resistant to cefepime. 10/18 K. pneumoniae isolates were resistant to multiple drugs with 7 isolates resistant to four or more antibiotics. fosA, oqxA, and oqxB were found in a majority of isolates, 14/18 (78%), 17/18 (94%), and 17/18 (94%) respectively (Supplementary Fig. 1D). Together, these data show that while AR Enterobacterales were recovered from fecal specimens and can be a cause of blood stream infection in patients in the ICUs, these isolates were rarely found on surfaces, with no clear relationships between source of isolation and MLST, building, or antibiotic resistance. This suggests patient fecal contamination of sampled surfaces in these ICUs was rare and did not lead to ARO reservoir formation.



Fig. 3 Timing, identity, and phylogenetics of *Stenotrophomonas* **isolates. a** Ratio of *Stenotrophomonas* isolates to all isolates across all time points (n = 128 *Stenotrophomonas* isolates). Error bars indicate standard error of intensive care unit (ICU) rooms. Red bars indicate collection timing of *Stenotrophomonas* blood culture isolates. **b** Identity of all collected *Stenotrophomonas* genomes by >95% average nucleotide identity (ANI) to reference genome by sample collection type (n = 128 isolates). Other Surface indicates all other surface/water genomes apart from in-room and bathroom sink drain. All genomes were identified as *Stenotrophomonas* by MASH. *Stenotrophomonas* various genomospecies includes all different genomospecies that did not share >95% ANI with a reference genome. **c** Time point mapping of shared *S. maltophilia* MLST groups by sample collection location. **d** Cladogram built from a core genome alignment of *S. maltophilia* genomes. Branches with less than 80% bootstrap support are collapsed. Branches with bootstrap values between 80-95% are labeled. BP before patient and staff move-in.

Stenotrophomonas maltophilia strains are found persistently across one year of sampling in single ICU rooms. While S. maltophilia is predominantly found in environmental water sources, the species is an emerging pathogen associated with HAIs, particularly in immunocompromised patients; these infections are associated with substantial case fatality rates⁶⁴, primarily because of the intrinsic antimicrobial resistance of this microorganism and the vulnerable patient population that it affects. Stenotrophomonas spp. were isolated from every week sampled, except for the first week of collection in the old ICU, although the ratio of Stenotrophomonas isolates to all collected isolates varied over time (Fig. 3a). Among 128 isolates identified as S. maltophilia by MALDI-TOF MS, ANI species identification and MASH genus identification typed them as 54 S. maltophilia isolates, 1 S. lactiubi, and 53 Stenotrophomonas spp. (not otherwise specified) in 9 genomospecies groupings (Fig. 4b). When we compared MLST and core genome phylogeny of S. maltophilia isolates, we found that sequence type and phylogenetic clades were not shared across ICUs (Fig. 3c, d, Supplementary Fig. 2). Only two sequence types were identified on the same respective surface over multiple weeks, suggesting that these surfaces acted as reservoirs (Fig. 3c, blue sequence types). S. maltophilia of ST27 was found 9 times over 35 weeks, and S. maltophilia of ST1 was found 13 times over the course of a year (56 weeks), including

before patient and staff occupancy (Fig. 3c). Both sequence types remained in the same room, with no evidence of crossover between rooms in the new ICU (Fig. 3c). Phenotypic susceptibly demonstrated no isolates with trimethoprim-sulfamethoxazole or minocycline resistance, one isolate with levofloxacin resistance, and 34/54 isolates with colistin resistance.

P. aeruginosa were diverse and found persistently across one year of sampling and in all 6 new ICU rooms. While commonly found in the environment, Pseudomonas spp. have a long history of causing HAIs^{60,70-73}. Studies have shown that P. aeruginosa reservoirs established in hospital built environments can lead to infections and outbreaks^{74–78}. However, it is unclear when these reservoirs became established, relative to patient or staff occupancy of the healthcare environment, and how pervasively Pseudomonas spp. may colonize ICU surfaces. We recovered more Pseudomonas spp. isolates than any other genus during our collections (Fig. 2b). MALDI-TOF MS identified 283 Pseudomonas spp. isolates. Pseudomonas spp., and particularly P. aeruginosa, isolates were collected at every time point in the study period, including before patient occupancy (Fig. 4a, b). After ANI species identification and MASH genus identification, we found 155 P. aeruginosa isolates, 71 Pseudomonas spp. isolates in 13



Fig. 4 Timing, identity, and phylogenetics of *Pseudomonas* **spp. isolates. a** Ratio of *Pseudomonas* **spp. to all isolates** across all time points (n = 283 *Pseudomonas* isolates). Error bars indicate standard error. Red bars indicate collection timing of *Pseudomonas aeruginosa* blood culture isolates. **b** Ratio of *P. aeruginosa* to all isolates across all time points (n = 155 *P. aeruginosa* isolates). Error bars indicate standard error. **c** Identity of all collected *Pseudomonas* **spp**. genomes by >95% ANI to reference genome by sample collection type. Other indicates all other surface/water genomes apart from in-room and bathroom sink drain. All genomes were identified as *Pseudomonas* spp. by MASH. *Pseudomonas* various genomospecies includes all different genomospecies that did not share >95% ANI with a reference genome. **d** Cladogram from a core genome alignment of *P. aeruginosa* genomes. Branches with less than 80% bootstrap support are collapsed. Branches with bootstrap values between 80–95% are labeled. Reference *P. aeruginosa* genomes included antibiotic-resistant (AR) isolates, clinical isolates, and environmental isolates. Reference MLST is included if it shares a MLST with collected isolates. **e** Time point mapping of top 8 MLST *P. aeruginosa* groups by sample collection location. BP before patient and staff move-in.

genomospecies groupings, and 54 isolates from other *Pseudo-monas* species (Fig. 4c). Most *P. aeruginosa* isolates were from surface samples (80%); 11% were from fecal samples; and 9% were from blood cultures (Fig. 4c). We did not find overlap between any other *Pseudomonas* spp. isolates from patient blood cultures and environmental samples.

When we compared MLST and core genome phylogeny of *P. aeruginosa* isolates, we find that isolates from different ICUs fall into different clades and strain types (Fig. 4d, e, Supplementary Fig. 3). To understand the genomic context of *P. aeruginosa* isolates, we compared the genomes of isolates recovered from surface and patient sampling with 172 reference *P. aeruginosa* genomes downloaded from NCBI (Supplementary Data 3). Reference genomes were phylogenetically diverse and fell into 3 categories: (i) isolates from clinical infections, (ii) AR isolates

from the CDC with known antibiotic resistance, and (iii) environmental isolates that had been collected from water and waste projects. The isolates we collected from both the old and new ICUs spanned most of the diversity of *P. aeruginosa* with no distinct clustering between collected ICU surface isolates and environmental, clinical, or AR isolates (Fig. 4d). Although there were no distinct clades based on isolate building or surface source, we do find that our isolates form a number of clades with highly-related surface isolates (Fig. 4d). These frequently corresponded with sequence type. There were two cases of overlap in sequence type between the old and the new building (Fig. 4e). ST17 was found in sink drains in both the old and new ICU and found in a blood culture in the new ICU. ST170 was found in surface samples in the old ICU and a patient fecal sample in the new ICU (Fig. 4e). Notably, *P. aeruginosa* of ST1894 was recovered from the same sink drain beginning before patient occupancy and continuing through for the full year of collection in the new ICU. This repeated isolation of ST1894 suggests that it may have established a continuous reservoir in this room in the new ICU. Furthermore, isolates of *P. aeruginosa* of ST1894 were also recovered from sink drains in all 6 sampled ICU rooms and were found across 5 or more time points in 5/6 sampled ICU rooms (Fig. 4e), suggesting this colonization and persistence is more widespread. Finally, we found that 3 blood culture isolates (3.7% of all blood culture isolates tested) also belonged to ST1894, which prompted a higher resolution comparative analysis of all ST1894 strains, due to its potential to contaminate the environment and be associated with bloodstream infections.

Antibiotic resistance in P. aeruginosa isolates varies between the two ICUs. To determine if antibiotic resistance profiles vary by location, we determined phenotypic susceptibility using antibiotic susceptibility testing (AST) and identified ARGs in assembled genomes using Resfinder⁵³. P. aeruginosa are defined as AROs because of their intrinsic resistance to many aminoglycosides, tetracyclines, β -lactams, and quinolones;^{60,79} we performed ASTs for 14 antibiotics for all Pseudomonas isolates to measure acquired resistances to β-lactams, cephalosporins, carbapenems, penicillins, fluoroquinolones, aminoglycosides, and polymyxins. AST profiles were similar across P. aeruginosa of the same sequence type (Fig. 5). P. aeruginosa isolates of ST1894 were largely not resistant to the antibiotics tested. P. aeruginosa isolates of ST282 were resistant to meropenem (11/15) and gentamicin (15/15). P. aeruginosa isolates of ST308 were resistant to meropenem (6/8), imipenem (5/8), ciprofloxacin (8/8), levofloxacin (8/8), and gentamicin (8/8). As different sequence types dominated the different ICUs and resistance profiles were similar across sequence types, we found trends in resistance to be different between the two ICUs with isolates from the old ICU

having a higher percentage of resistance to meropenem and imipenem than *P. aeruginosa* isolates from the new ICU (new ICU: 7% imipenem, 7% meropenem and old ICU: 40% imipenem, 55% meropenem) (Fig. 5).

Much like the AST profiles, the ARG profiles also appeared to be linked to sequence type (Fig. 5). Nearly all *P. aeruginosa* isolates carried the resistance genes aph(3')-IIb (153/155), blaPAO(154/155), catB7 (151/155), and fosA (155/155). Isolates from ST282 were the only identified to contain the aminoglycoside resistance genes aac(6')-IIb (15/15) and aadA1b (15/15), which could explain the phenotypic resistance to gentamicin. All isolates from ST 1894 carried the β -lactam resistance gene blaOXA-396 (52/52), while it was less common in other sequence types (35/ 103). *P. aeruginosa* is also capable of developing resistance to carbapenems (meropenem, imipenem), fluoroquinolones (ciprofloxacin, levofloxacin), and aminoglycosides (gentamicin) by chromosomal point mutations, rather than acquisition of ARGs^{80,81}.

P. aeruginosa Group 1 strain was found across 1 year of sampling and in both environmental and patient samples. While MLST has been used previously to describe strains and outbreaks, it is limited to a small number of genes or alleles and does not enable genome-resolved understanding of strain relatedness. Accordingly, here we utilized WGS data for each P. aeruginosa isolate to perform an in-depth analysis of similarity across genomes. We calculated pairwise SNP distances by mapping quality filtered short-reads from all P. aeruginosa isolates to a high-quality, long-read sequencing-assembled genome of the first temporal occurrence of ST1894, with a mean of 89.8% of reads mapped to the genome. We then used a grouping technique on *P. aeruginosa* isolates to find fully reciprocal groups⁸². We compared pairwise SNP distances between P. aeruginosa isolate pairs and iterated through each unique SNP distance cutoff to filter the isolate pairwise network list (Fig. 6a, b). For each SNP



Fig. 5 Phenotypic and genotypic antibiotic resistance of collected *P. aeruginosa* isolates. Phylogenetic tree is from a core genome alignment. Phenotypic resistance determined by antibiotic susceptibility testing (AST). Genotypic resistance determined by Resfinder⁵³.



Fig. 6 Highly-related genomic groups of *P. aeruginosa* across locations and time. a Histogram of pairwise single nucleotide polymorphism (SNP) distances between *P. aeruginosa* genomes indicate three modes of pairwise distances. The first corresponds to highly-related genomic groups. We define group SNP threshold as pairwise distances that fall before 2743 (gray dashed line). b Zoomed in histogram of pairwise SNP distances between *P. aeruginosa* genome with a cut off at 3000 SNPs show only highly-related genomic groups. c Max groupings by SNP cut off show pairwise groups plateau at 2743 SNPs. d Number of isolates per highly-related genomic group. Other surfaces includes all other surfaces that are not in-room or bathroom sink drain. e First two components of principal component analysis (PCA) of the accessory genome of all *P. aeruginosa* genomes. Black circle encloses all Group 1 *P. aeruginosa* genomes. **f** Time point mapping of top 4 *P. aeruginosa* highly-related groups and highly-related groups that shared isolates between patient and surface samplings. **g** Time-measured phylogenetic analysis consensus tree of n = 48 Group 1 *P. aeruginosa* isolates depicted using DensiTree v2.2.7⁵⁹. Nodes labeled with black circles. Node 1 marks the main clade with a time since most recent common ancestor (TMRCA) of 778 days. BP before patient and staff move-in.

cutoff, we determined the number of complete subgraph groups, defined by each node in the group was connected to every other node in the group, and isolates per group. The number of P. aeruginosa groups rose initially from 3 to 18 groups as SNP distances increased from 0 to 377 SNPs. After a peak at 756 SNPs with 20 groups, the total number of groups slowly decreased to a plateau of 14 groups at 2743 SNPs (Fig. 6c). From this, we determined an appropriate SNP cutoff that separated closelyrelated isolates from other groups was 2743. Using this definition, P. aeruginosa isolates fell into 14 groups, with the largest group (Group 1) including 53 isolates (Fig. 6d). Only three groups had isolates that spanned patient and environmental isolates: Group 1, Group 6, and Group 12. Group 1 had no more than 11 SNPs between isolates and included isolates from blood cultures and environmental samples. 52/53 of the isolates in Group 1 were ST1894, and the remaining isolate was unidentified but had 5/6 alleles identical to ST1894. The isolates in this group persistently and pervasively colonized new ICU sink drains and were cultured from sink drains 49 times across 56 weeks (Fig. 6f). Aside from sink drains, Group 1 was also found in 3 patient blood cultures, 1 of which was isolated from a different ward in the same building (Fig. 6f). 1 isolate from Group 1 was isolated from the gown and glove personal protective equipment box located just outside the room. All isolates within this group were within 11 short-read SNPs of each other. Group 2 (ST17) was found once in a sink drain in the old ICU, 7 times in the bathroom sink drain of Room 5 in the new ICU, and once in a blood culture isolate. Group 12 (ST241) was found once in a sink drain in the new ICU, and once in a blood culture isolate. This highlights 3 instances where a sink drain isolate was found within the same genome-resolved group as a blood culture isolate of a patient in the ICU.

When we compared the accessory genomes of our cultured isolates and reference *P. aeruginosa* genomes, we found a mean of 4018 (range 3221–5003) accessory genes per genome. Group 1 isolates have a mean of 3947 (range 3885–4022) accessory genes, suggesting average accessory genome size. To compare variation in accessory genomes across *P. aeruginosa* we used a principal component analysis (PCA). We found distinct clustering between Group 1 isolates and the rest of the *P. aeruginosa* isolates (PERMANOVA: p < 0.001) (Fig. 6e). There were 36 accessory genes with high loading scores on PC1 that are unique to Group 1, of which only 7 could be characterized by EggNOG (Supplementary Data 4, Supplementary Data 5)⁵⁰.

P. aeruginosa Group 1 isolate lineages clustered by room. SNP analysis from short reads allows us to accurately estimate genomic relatedness and group highly-related genomes, but obtaining fully-resolved genomes is necessary to identify transmission and reservoir persistence in sink drains. Our short-read data indicated <11 SNPs between Group 1 isolates, which is well within previously established probable transmission for P. aeruginosa⁸³. Indeed, our phylogenomic tree building grouped isolates similarly to this analysis (Fig. 4d, Supplementary Fig. 3). To investigate reservoir formation of P. aeruginosa Group 1 isolates over time at higher genomic resolution, we obtained long-read sequencing data for the 53 isolates in Group 1. We created hybrid assemblies of each genome (assemblies had between 1-11 and an average of 4.32 contigs) and found the core genome to consist of 4863 genes out of 9714 total genes. By examining the accessory genome, we identified 4 additional isolates that were responsible for a large portion of the accessory genome and removed them from analysis, as they were unlikely to be part of the same lineage. The remaining 49 Group 1 isolates consisted of 5964 core genes from 6986 total genes.

In our time-measured phylogenetic analysis using BEAST⁵⁴, we created a consensus tree of estimated time since most recent common ancestor (TMRCA) using the Group 1 core genome (Supplementary Fig. 4). As we do not have isolates collected this far back, confidence in branch divisions is low (Supplementary Fig. 5) and the TMRCA of 8034 days was largely driven by one isolate, which was removed from further analysis. The remaining isolates formed a consensus tree (Fig. 6g) with a TMRCA of 2752 days with a 95% highest posterior density interval (HPD) of 1523–4362 days. 38/48 of these isolates were taken from the inroom sink drains; 7 were found in the sink drains from the attached bathroom; and 3 were isolated from blood infections.

The majority (40/48) of these isolates are contained under Node 1 and have a TMRCA of 778 days with a 95% HPD of 488-1122 days. This clade displays 3 unique evolutionary patterns. Descending from Node 2, eight isolates cluster together. 7/8 of these isolates were taken from the in-room sink drains in Room 1 within the first 2 weeks of the study, suggesting the diversity shown represents heterogeneity within a given sink rather than in-room evolution. We also found two likely instances of within room evolution, marked by Nodes 3 and 4. These isolates branch off from one lineage as time progresses, suggesting an evolving, single reservoir. Outside of these main features, the remaining isolates under Node 1 were from mixed rooms and collection weeks with low confidence in the branching (Supplementary Fig. 4). The commingling and low confidence in branching suggests strain exchange between rooms of a common pandemic strain throughout the sampling period.

Interestingly, one of the blood isolates clustered with an environmental isolate within the clade under Node 1, but the other two were further removed from the bulk of Group 1. While all isolates were collected from patients in this ward at some point during their stay, this isolate was collected two days before the patient moved into Room 5. The other two blood isolates were further removed from Node 1 environmental isolates. However, there are possibly different evolutionary pressures within each sample type, which could drive different mutation rates. The overall close relation of the blood and surface isolates implies direct correlation within the duration of the study period and potentially presents a great risk to patient safety.

Discussion

The process of ARO reservoir colonization of the hospital built environment is dependent on complex interactions, and transmission events to vulnerable patients are not well understood^{27,84,85}. In this study we investigated the microbiologic changes in a new SCT ICU before and after patient or staff occupancy and tracked ARO strains cross ICU surfaces and patients. We identified a mechanism of ARO colonization development that occurred prior to patient or staff move-in, which could promote the necessity of future surveillance investigations. We compared these colonization patterns to equivalent microbial sampling in the corresponding old SCT ICU during its final month of occupancy, before patients and staff moved from there to the new ICU, to obtain a unique comparison of distinct hospital built environments following relocation of patients and hospital staff. We found ARO reservoirs were rare on most hospital surfaces apart from sink drains. Non-sink drain surfaces showed no difference in ARO burden between the two buildings, but sink drains in the old ICU had a significantly higher burden than those in the new ICU. Further, reservoir colonization and transmission varied by taxa and between buildings, with evidence in P. aeruginosa of shared strains across multiple sinks and human clinical infections in the new ICU.

Recent studies have focused on better understanding and characterizing the hospital microbiome using metagenomics^{11–15}. These characterizations find correlations between samples of hospital surfaces, patients, and staff, particularly in skin- and gutassociated taxa such as Enterobacterales and Staphylococcus, suggesting the microbiomes of humans and the hospital built environment influence each other¹¹⁻¹⁵. Further, strain tracking using metagenomic analyses indicates that similar strains may be present on surfaces over time, suggesting potential reservoir colonization on surfaces^{11,13}. Building on these studies, we focused on high-resolution, temporal, genomic and phenotypic investigation of viable AR strains which colonize or infect surfaces and patients in ICUs. We found that AROs isolated from patient stools were rarely found on ICU surfaces, and with the exception of sink drains, we do not find persistent reservoir colonization of most ICU surfaces. In contrast we found multiple instances of ARO reservoir colonization of ICU sink drains, with highly-related strains of these AROs also recovered from patient blood cultures.

We found AROs more frequently in sink drains in the old ICU compared to the new ICU. There are many possible reasons for these differences, including: building material and layout differences, water sources, natural history, and extended time for establishment and accumulation of AROs^{86–89}. In the new ICU, AROs were found before patient or staff occupancy, and ARO burden in both sink drains and other surfaces did not significantly change after 1 year of patient occupancy. This baseline level of ARO burden in an ICU suggests that patients are not the primary source of AROs found on surfaces nor do they cause significant increases in ARO burden during the first year of ICU establishment. Further, these results have important implications for remediation strategies that involve removing or rebuilding infrastructure and suggests such strategies may not always be successful.

When comparing ICU room surfaces, we found AROs more frequently on sink drains and rarely on any other ICU room surface. While studies in low to medium income countries have found high ARO burdens on hospital surfaces, our results are consistent with other studies in the United States (US) that have found low ARO burden on ICU surfaces and high ARO burden in sink drains^{82,90}. While it is possible that our sampling methods may miss some AROs, the sparsity and inconsistency of AROs on surfaces suggests that most surfaces other than sink drains are not acting as persistent reservoirs for AROs. It may also be that some AROs do not survive well on dry ICU surfaces where they cannot easily form biofilms^{88,90}. However, other studies have found ARO colonization on these types of surfaces for long periods, suggesting that colonization is possible^{82,91-93}. Instead, high standards of cleaning, self-disinfecting equipment, and special training in high income countries such as the US may be effective at removing and limiting ARO reservoirs on most commonlytouched surfaces^{91,94,95}. While national standards and studies have suggested protocols for cleaning many hospital surfaces⁹⁵, there are no standardized protocols for cleaning sink drains. This may lead to variable and inconsistent decontamination of these areas compared to other commonly-touched surface areas. Further, sink drains are often difficult to clean as liquid disinfectant is less effective when poured down the drain without coating the drain surface, and the drains are often covered by a drain cover and cannot easily be wiped down or scrubbed^{96,97}.

Reservoir colonization by AROs in sink drains appears to be specific to particular taxa. While we cultured a wide diversity of AROs from sink drains, only two species had strains that formed reservoirs in sink drains: *S. maltophilia* and *P. aeruginosa*. These results corroborate previous work identifying *Pseudomonas* spp. and *Stenotrophomonas* spp. as capable of long-term colonization of sink drains^{74–77}. In contrast, we did not find evidence of persistent colonization of sink drains by Enterobacterales species, which have commonly been associated with hospital built environment outbreaks^{17,28,88,98–100}. It is possible these organisms were present but weren't isolated because they were not resistant to the antibiotics used in selective culturing.

S. maltophilia is an environmental organism that is emerging as a serious concern for HAIs and other infections⁶⁴. For our purposes, we defined reservoirs to mean surfaces where at least 2 isolates from the same sequence type were isolated from consecutive samplings. We found reservoirs of S. maltophilia in at least 3 sink drains. S. maltophilia ST1 established reservoirs in two surfaces of the same ICU room, suggesting a similar source or the spread of one strain type to a different location. However, we find little evidence of strain transfer to sink drains in other rooms in the same ICU, and no evidence of transmission to patients. In fact, while we found 3 Stenotrophomonas isolates in blood cultures, when we used ANI to identify species, none of these were identified as S. maltophilia. This may have broad clinical applications as poor identification of blood isolates could potentially lead to inappropriate treatment. However, even though we find no evidence of transmission of S. maltophilia sink strains to patients, since S. maltophilia has been shown to be a pathogen in immunocompromised patients, it is still important to identify methods to remove sink drain reservoirs of these organisms.

P. aeruginosa has long been characterized as an opportunistic pathogen that inhabits environmental sources, particularly water sources, as well as the human $gut^{60,70-73}$. When compared to diverse *P. aeruginosa* genomes from other studies, we found no distinct clustering with environmental or clinical isolates, suggesting that our isolates are not coming from a strictly environmental strain pool. Instead, the strains we characterized were phylogenetically diverse, indicating that the adaptations necessary to survive in sink drains in the ICU are not restricted to a single clade. Further, there was limited apparent transfer of surface isolates between ICUs as patients and staff moved from one location to the other, as strains were unique between surfaces in the old and new ICU.

Remarkably, the genomic diversity of P. aeruginosa isolates from ST1894 in sink drains was incredibly low, even after one year or sampling. P. aeruginosa ST1894 was first described in 2014 in a cystic fibrosis patient in Spain (Isolate RC19, id:2398)⁴⁴. Since only the MLST was done for this isolate and not WGS, it is impossible to determine if this isolate and our ST1894 isolates have similar ancestry. The ST1894 isolates from our study are not only capable of surviving well in sink drains but also of colonizing multiple sink drains; our collection scheme documents Group 1 P. aeruginosa of ST1894 first being cultured from a single room, but after 17 weeks of sample collection, was found in all six ICU rooms samples. Our initial short read-based WGS approach provided the resolution to cluster isolates into groups based on whole-genome SNP distances, which has been the mainstay for transmission dynamics up until this point^{83,101,102}. However, our long-read sequencing analysis elucidated the more nuanced relationships necessary for transmission and reservoir colonization dynamics. Specifically, our phylogenetic analysis with highquality hybrid assemblies indicates key cases of a ST1894 strain inhabiting one sink drain before patients move into the hospital, and then spreading and exchanging between all rooms sampled. Our sampling illuminated the diversity and evolution of this lineage across time and space during the course of the study period. We also found evidence of 3 instances where this strain was found in blood cultures from hospitalized patients, highlighting ST1894 as an urgent threat to this healthcare facility and associated immunocompromised patients. The bias for these isolates originating from the in-room sink drains rather than the

bathroom sink drains also suggests that the water source system, which is common to all drains, is not a likely source of reservoir contamination. This, in association with the patient sample that was collected outside the sampling ward, lead us to suspect this strain may be more widespread in this healthcare system than our sampling area. Our genomic analyses indicate that *P. aeruginosa* ST1894 has a very unique accessory genome compared to other *P. aeruginosa*, thus leaving a long list of candidate genes that might explain its prevalence in sink drains. Further investigation into these genes and other similar strains will help us better understand the genomic evolution that might have allowed for its environmental pathogenicity.

Globally, antibiotic resistance in P. aeruginosa isolates is a growing concern, with infection mortality rates of 33-71% in carbapenem-resistant infections⁸¹. P. aeruginosa is capable of both intrinsic chromosomal modifications and acquisition of mobile ARGs that encode resistance to all classes of antibiotics currently used in P. aeruginosa treatment. However, carbapenem resistance in P. aeruginosa has only been acquired through the acquisition of mobile ARGs, most commonly metallo- β -lactamases (MBLs) and are typically encoded on plasmids, integrons, and mobile cassettes⁸¹. In general, carbapenem resistance was rare in P. aeruginosa isolates collected in the new ICU, while it was common in P. aeruginosa isolates collected in the old ICU. P. aeruginosa ST1894 was generally susceptible to the suite of antibiotics we tested against, with only 2 instances of resistance observed. Fortunately, this means there are currently a number of viable antibiotic treatment options against the existing reservoirs of ST1894 in our healthcare system. However, the presence of other Pseudomonas spp. with much higher AR burdens in this same hospital environment, and the known ease of resistance transmission in Pseudomonas spp., emphasizes the risk that this widely disseminated ST1894 reservoir could evolve into a greater ARO threat.

Despite our success in identifying multiple reservoirs with our current methods, it is plausible that we are under sampling the genomic diversity and persistent colonization through the cross section of time points sampled. For example, in our identification of reservoirs by Group 1 of *P. aeruginosa* (Fig. 6f), we believe the strain was likely still present even when it appears to skip certain weeks. Even with selection of multiple isolates per selective plate, further work could improve these methods, such as a metagenomics based approach, and reveal additional reservoirs.

It is intriguing that many AROs were found in sink drains even prior to patient relocation to this unit. Previous work has suggested sources of contamination such as patient or hospital staff carriage of *P. aeruginosa*¹⁰³, or diffusion through water pipes^{103,104}, but these don't address contamination identified prior to patient or staff move-in. Other studies have identified water contamination as a potential source^{75,76,105}, but our sampling did not indicate water as the source of these AROs. Further research is necessary to understand the origins of the strains. Regardless of their origins, these findings highlight the need for a more thorough decontamination procedures, both during the terminal clean and regular operation of ICU facilities.

In conclusion, our investigation of ARO reservoirs allowed us to assess and compare models of colonization and transmission in an old and new hospital built environment with the same patient and staff populations, including before and after patient or staff occupancy. Our approach of selective microbiologic culture combined with WGS analyses provide for a detailed analysis of ARO variation across one year of sampling in an SCT ICU. Together these data provide a high-resolution characterization of AROs in the hospital built environment, highlighting that SCT ICU sink drains are a major reservoir for AROs with direct links to patient infections. Most pressingly, the surprisingly rapid development of *P. aeruginosa* colonization and association with patient infections emphasizes the need for future work to decrease the spread of AROs in hospital built environments, completed by efforts towards decolonizing and eliminating sink drain ARO reservoirs.

Data availability

All genomic reads generated during and/or analyzed during the current study are available under BioProject PRJNA741123. Other source data for the main figures can be found in Supplementary Data 1–5.

Received: 5 November 2021; Accepted: 6 May 2022; Published online: 01 June 2022

References

- Al-Tawfiq, J. A. & Tambyah, P. A. Healthcare associated infections (HAI) perspectives. J. Infect. Public Health 7, 339–344 (2014).
- Ariza-Heredia, E. J. & Chemaly, R. F. Update on infection control practices in cancer hospitals. CA Cancer J. Clin. 68, 340–355 (2018).
- Burnham, J. P., Olsen, M. A. & Kollef, M. H. Re-estimating annual deaths due to multidrug-resistant organism infections. *Infect. Control Hosp. Epidemiol.* 40, 112–113 (2019).
- Rosenthal, V. D., Guzman, S., Migone, O. & Crnich, C. J. The attributable cost, length of hospital stay, and mortality of central line-associated bloodstream infection in intensive care departments in Argentina: a prospective, matched analysis. *Am. J. Infect. Control* **31**, 475–480 (2003).
- Giraldi, G., Montesano, M., Sandorfi, F., Iachini, M. & Orsi, G. B. Excess length of hospital stay due to healthcare acquired infections: methodologies evaluation. *Ann. Ig* 31, 507–516 (2019).
- Serafim, R. B., Povoa, P., Souza-Dantas, V., Kalil, A. C. & Salluh, J. I. F. Clinical course and outcomes of critically ill patients with COVID-19 infection: a systematic review. *Clin. Microbiol. Infect.* 27, 47–54 (2021).
- Yilmaz, C. & Ozcengiz, G. Antibiotics: Pharmacokinetics, toxicity, resistance and multidrug efflux pumps. *Biochem. Pharmacol.* 133, 43–62 (2017).
- Weiner, L. M. et al. Antimicrobial-resistant pathogens associated with healthcare-associated infections: summary of data reported to the National Healthcare Safety Network at the Centers for Disease Control and Prevention, 2011-2014. *Infect. Control Hosp. Epidemiol.* 37, 1288–1301 (2016).
- Hatier, R. & Grignon, G. Ultrastructural study of the Sertoli cell and the limiting membrane in the seminiferous tubule of the adult cryptorchid rat. *Virchows Arch. B Cell Pathol. Incl. Mol. Pathol.* 52, 305–318 (1986).
- Weber, D. J., Anderson, D. & Rutala, W. A. The role of the surface environment in healthcare-associated infections. *Curr. Opin Infect. Dis.* 26, 338–344 (2013).
- Brooks, B. et al. Strain-resolved analysis of hospital rooms and infants reveals overlap between the human and room microbiome. *Nat. Commun.* 8, 1814 (2017).
- Lax, S. et al. Bacterial colonization and succession in a newly opened hospital. Sci. Transl. Med. 9, https://doi.org/10.1126/scitranslmed.aah6500 (2017).
- ElRakaiby, M. T., Gamal-Eldin, S., Amin, M. A. & Aziz, R. K. Hospital microbiome variations as analyzed by high-throughput sequencing. *OMICS* 23, 426–438 (2019).
- Rampelotto, P. H., Sereia, A. F. R., de Oliveira, L. F. V. & Margis, R. Exploring the hospital microbiome by high-resolution 16S rRNA profiling. *Int. J. Mol. Sci.* 20, https://doi.org/10.3390/ijms20123099 (2019).
- Yano, R. et al. Diversity changes of microbial communities into hospital surface environments. J. Infect. Chemother. 23, 439–445 (2017).
- Mihara, T. et al. Secondary in-hospital epidemiological investigation after an outbreak of Pseudomonas aeruginosa ST357. J. Infect. Chemother. 26, 257–265 (2020).
- 17. Snitkin, E. S. et al. Tracking a hospital outbreak of carbapenem-resistant Klebsiella pneumoniae with whole-genome sequencing. *Sci. Transl. Med.* **4**, 148ra116 (2012).
- Gotoh, Y. et al. Multi-step genomic dissection of a suspected intra-hospital Helicobacter cinaedi outbreak. *Microb. Genom.* 4, https://doi.org/10.1099/ mgen.0.000236 (2018).
- Fernando, S. A., Gray, T. J. & Gottlieb, T. Healthcare-acquired infections: prevention strategies. *Intern. Med. J.* 47, 1341–1351 (2017).
- Abreu, A. C., Tavares, R. R., Borges, A., Mergulhao, F. & Simoes, M. Current and emergent strategies for disinfection of hospital environments. *J. Antimicrob. Chemother.* 68, 2718–2732 (2013).
- Facciola, A. et al. The role of the hospital environment in the healthcareassociated infections: a general review of the literature. *Eur. Rev. Med. Pharmacol. Sci.* 23, 1266–1278 (2019).
- 22. Weber, D. J., Rutala, W. A., Miller, M. B., Huslage, K. & Sickbert-Bennett, E. Role of hospital surfaces in the transmission of emerging health care-

ARTICLE

associated pathogens: norovirus, Clostridium difficile, and Acinetobacter species. Am. J. Infect. Control 38, S25–S33 (2010).

- Dancer, S. J. The role of environmental cleaning in the control of hospitalacquired infection. J. Hosp. Infect. 73, 378–385 (2009).
- 24. Costa, D. et al. Nosocomial outbreak of Pseudomonas aeruginosa associated with a drinking water fountain. J. Hosp. Infect. 91, 271–274 (2015).
- Schneider, H. et al. Pseudomonas aeruginosa outbreak in a pediatric oncology care unit caused by an errant water jet into contaminated siphons. *Pediatr. Infect. Dis. J.* **31**, 648–650 (2012).
- Loveday, H. P. et al. Association between healthcare water systems and Pseudomonas aeruginosa infections: a rapid systematic review. J. Hosp. Infect. 86, 7–15 (2014).
- Volling, C. et al. Are sink drainage systems a reservoir for hospital-acquired gammaproteobacteria colonization and infection? A systematic review. Open Forum Infect. Dis. 8, ofaa590 (2021).
- Decraene, V. et al. A large, refractory nosocomial outbreak of klebsiella pneumoniae carbapenemase-producing Escherichia coli demonstrates carbapenemase gene outbreaks involving sink sites require novel approaches to infection control. *Antimicrob. Agents Chemother.* 62, https://doi.org/10. 1128/AAC.01689-18 (2018).
- Sethi, A. K., Al-Nassir, W. N., Nerandzic, M. M., Bobulsky, G. S. & Donskey, C. J. Persistence of skin contamination and environmental shedding of Clostridium difficile during and after treatment of C. difficile infection. *Infect. Control Hosp. Epidemiol.* 31, 21–27 (2010).
- Hakim, H. et al. Gut microbiome composition predicts infection risk during chemotherapy in children with acute lymphoblastic leukemia. *Clin. Infect Dis.* 67, 541–548 (2018).
- Kuo, F. C. et al. Bloodstream infections in pediatric patients with acute leukemia: emphasis on gram-negative bacteria infections. J. Microbiol. Immunol. Infect. 50, 507–513 (2017).
- Atkins, S. & He, F. Chemotherapy and beyond: infections in the era of old and new treatments for hematologic malignancies. *Infect. Dis. Clin. North Am.* 33, 289–309 (2019).
- CLSI. M100 performance standards for antimicrobial susceptibility testing (CLSI, 2021).
- 34. Bolger, A. M., Lohse, M. & Usadel, B. Trimmomatic: a flexible trimmer for Illumina sequence data. *Bioinformatics* **30**, 2114–2120 (2014).
- Bankevich, A. et al. SPAdes: a new genome assembly algorithm and its applications to single-cell sequencing. J. Comput. Biol. 19, 455–477 (2012).
- Gurevich, A., Saveliev, V., Vyahhi, N. & Tesler, G. QUAST: quality assessment tool for genome assemblies. *Bioinformatics* 29, 1072–1075 (2013).
- Parks, D. H., Imelfort, M., Skennerton, C. T., Hugenholtz, P. & Tyson, G. W. CheckM: assessing the quality of microbial genomes recovered from isolates, single cells, and metagenomes. *Genome Res.* 25, 1043–1055 (2015).
- Seemann, T. Prokka: rapid prokaryotic genome annotation. *Bioinformatics* 30, 2068–2069 (2014).
- 39. Wick, R. & Menzel, P. Filtlong. https://github.com/rrwick/Filtlong (2021).
- Kolmogorov, M., Yuan, J., Lin, Y. & Pevzner, P. A. Assembly of long, errorprone reads using repeat graphs. *Nat. Biotechnol.* 37, 540–546 (2019).
- 41. Walker, B. J. et al. Pilon: an integrated tool for comprehensive microbial variant detection and genome assembly improvement. *PLoS One* **9**, e112963 (2014).
- Marcais, G. et al. MUMmer4: A fast and versatile genome alignment system. PLoS Comput. Biol. 14, e1005944 (2018).
- 43. Ondov, B. D. et al. Mash: fast genome and metagenome distance estimation using MinHash. *Genome Biol.* **17**, 132 (2016).
- Jolley, K. A., Bray, J. E. & Maiden, M. C. J. Open-access bacterial population genomics: BIGSdb software, the PubMLST.org website and their applications. *Wellcome Open Res.* 3, 124 (2018).
- 45. Seemann, T. & da Silva, A. G. mlst. https://github.com/tseemann/mlst (2020).
- Page, A. J. et al. Roary: rapid large-scale prokaryote pan genome analysis. Bioinformatics 31, 3691–3693 (2015).
- Price, M. N., Dehal, P. S. & Arkin, A. P. FastTree: computing large minimum evolution trees with profiles instead of a distance matrix. *Mol. Biol. Evol.* 26, 1641–1650 (2009).
- Yu, G., Smith, D. K., Zhu, H., Guan, Y. & Lam, T. T.-Y. ggtree: an R package for visualization and annotation of phylogenetic trees with their covariates and other associated data. *Methods Ecol. Evol.* 8, 28–36 (2017).
- Yu, G., Lam, T. T.-Y., Zhu, H. & Guan, Y. Two methods for mapping and visualizing associated data on phylogeny using ggtree. *Mol. Biol. Evol.* 35, 3041–3043, (2018).
- Huerta-Cepas, J. et al. eggNOG 5.0: a hierarchical, functionally and phylogenetically annotated orthology resource based on 5090 organisms and 2502 viruses. *Nucleic Acids Res.* 47, D309–D314 (2019).
- Seemann, T. Snippy: rapid haploid variant calling and core genome alignment. https://github.com/tseemann/snippy (2021).
- 52. Csardi, G. & Nepusz, T. The igraph software package for complex network research. *InterJ. Complex Syst.* 2–8 (2006).

- Zankari, E. et al. Identification of acquired antimicrobial resistance genes. J. Antimicrob. Chemother. 67, 2640–2644 (2012).
- 54. Bouckaert, R. et al. BEAST 2.5: an advanced software platform for Bayesian evolutionary analysis. *PLoS Comput. Biol.* **15**, e1006650 (2019).
- Stecher, G., Tamura, K. & Kumar, S. Molecular Evolutionary Genetics Analysis (MEGA) for macOS. *Mol. Biol. Evol.* 37, 1237–1239 (2020).
- Barido-Sottani, J. et al. Taming the BEAST-A community teaching material resource for BEAST 2. Syst. Biol. 67, 170–174 (2018).
- Rambaut, A., Drummond, A. J., Xie, D., Baele, G. & Suchard, M. A. Posterior summarization in bayesian phylogenetics using tracer 1.7. Syst. Biol. 67, 901–904 (2018).
- Rambaut, A., Suchard, M., Nenarokov, S. & Klotzl, F. FigTree. https://github. com/rambaut/figtree/releases (2018).
- Bouckaert, R. & Heled, J. DensiTree 2: seeing trees through the forest. Preprint at https://www.biorxiv.org/content/10.1101/012401v1 (2014).
- Pang, Z., Raudonis, R., Glick, B. R., Lin, T. J. & Cheng, Z. Antibiotic resistance in Pseudomonas aeruginosa: mechanisms and alternative therapeutic strategies. *Biotechnol. Adv.* 37, 177–192 (2019).
- Sanchez, M. B., Hernandez, A. & Martinez, J. L. Stenotrophomonas maltophilia drug resistance. *Future Microbiol.* 4, 655–660 (2009).
- Spigaglia, P., Mastrantonio, P. & Barbanti, F. Antibiotic Resistances of Clostridium difficile. Adv. Exp. Med. Biol. 1050, 137–159 (2018).
- Berube, B. J., Rangel, S. M. & Hauser, A. R. Pseudomonas aeruginosa: breaking down barriers. *Curr. Genet.* 62, 109–113 (2016).
- Brooke, J. S. Stenotrophomonas maltophilia: an emerging global opportunistic pathogen. Clin. Microbiol. Rev. 25, 2–41 (2012).
- Jang, J. et al. Environmental Escherichia coli: ecology and public health implications-a review. J. Appl. Microbiol. 123, 570–581 (2017).
- Lindblom, A., Karami, N., Magnusson, T. & Ahren, C. Subsequent infection with extended-spectrum beta-lactamase-producing Enterobacteriaceae in patients with prior infection or fecal colonization. *Eur. J. Clin. Microbiol. Infect. Dis.* 37, 1491–1497 (2018).
- Karanika, S., Karantanos, T., Arvanitis, M., Grigoras, C. & Mylonakis, E. Fecal colonization with extended-spectrum beta-lactamase-producing enterobacteriaceae and risk factors among healthy individuals: a systematic review and metaanalysis. *Clin. Infect. Dis.* 63, 310–318 (2016).
- Bortolaia, V. et al. ResFinder 4.0 for predictions of phenotypes from genotypes. J. Antimicrob. Chemother. 75, 3491–3500 (2020).
- Zankari, E. et al. PointFinder: a novel web tool for WGS-based detection of antimicrobial resistance associated with chromosomal point mutations in bacterial pathogens. J. Antimicrob. Chemother. 72, 2764–2768 (2017).
- Ruiz-Garbajosa, P. & Canton, R. Epidemiology of antibiotic resistance in Pseudomonas aeruginosa. Implications for empiric and definitive therapy. *Rev. Esp. Quimioter.* **30**, 8–12 (2017).
- Cardozo, C., Rico, V., Aguero, D. & Soriano, A. Antibiotic selection in the treatment of acute invasive infections by Pseudomonas aeruginosa. *Rev. Esp. Quimioter.* 32, 32–34 (2019).
- Gonzalez-Olvera, E. M. et al. Antibiotic resistance, virulence factors and genotyping of Pseudomonas aeruginosa in public hospitals of northeastern Mexico. J. Infect. Dev. Ctries 13, 374–383 (2019).
- 73. CDC. Antibiotic resistance threats in the United States (CDC, 2019).
- Weng, M. K. et al. Outbreak investigation of Pseudomonas aeruginosa infections in a neonatal intensive care unit. Am. J. Infect. Control 47, 1148–1150 (2019).
- Bicking Kinsey, C. et al. Pseudomonas aeruginosa outbreak in a neonatal intensive care unit attributed to hospital tap water. *Infect. Control Hosp. Epidemiol.* 38, 801–808 (2017).
- Zhang, Y. et al. Bronchoscope-related Pseudomonas aeruginosa pseudooutbreak attributed to contaminated rinse water. Am. J. Infect. Control 48, 26–32 (2020).
- Walker, J. & Moore, G. Pseudomonas aeruginosa in hospital water systems: biofilms, guidelines, and practicalities. J. Hosp. Infect. 89, 324–327 (2015).
- Gbaguidi-Haore, H. et al. A bundle of measures to control an outbreak of Pseudomonas aeruginosa associated with P-trap contamination. *Infect. Control Hosp. Epidemiol.* **39**, 164–169 (2018).
- Hancock, R. E. & Speert, D. P. Antibiotic resistance in Pseudomonas aeruginosa: mechanisms and impact on treatment. *Drug Resist. Updat* 3, 247–255 (2000).
- Feng, Y., Jonker, M. J., Moustakas, I., Brul, S. & Ter Kuile, B. H. Dynamics of mutations during development of resistance by Pseudomonas aeruginosa against five antibiotics. *Antimicrob. Agents Chemother.* 60, 4229–4236 (2016).
- Bassetti, M., Vena, A., Croxatto, A., Righi, E. & Guery, B. How to manage Pseudomonas aeruginosa infections. *Drugs Context* 7, 212527 (2018).
- D'Souza, A. W. et al. Spatiotemporal dynamics of multidrug resistant bacteria on intensive care unit surfaces. *Nat. Commun.* 10, 4569 (2019).
- Bartell, J. A. et al. Omics-based tracking of Pseudomonas aeruginosa persistence in "eradicated" cystic fibrosis patients. *Eur. Respir. J.* 57, https:// doi.org/10.1183/13993003.00512-2020 (2021).

- Bloomfield, S. et al. Lesser-known or hidden reservoirs of infection and implications for adequate prevention strategies: where to look and what to look for. *GMS Hyg. Infect. Control* 10, Doc04 (2015).
- Capolongo, S., Stettimo, G. & Gola, M. Analysis of microorganisms in hospital environments and potential risks. *Indoor Air Qual. Healthcare Facilities*, 53–62 (2017).
- Julia, L. et al. Environmental reservoirs of nosocomial infection: imputation methods for linking clinical and environmental microbiological data to understand infection transmission. AMIA Annu. Symp. Proc. 2017, 1120–1129 (2017).
- Decker, B. K. & Palmore, T. N. The role of water in healthcare-associated infections. *Curr. Opin. Infect. Dis.* 26, 345–351 (2013).
- Kizny Gordon, A. E. et al. The hospital water environment as a reservoir for carbapenem-resistant organisms causing hospital-acquired infections-a systematic review of the literature. *Clin. Infect. Dis.* 64, 1435–1444 (2017).
- Chia, P. Y. et al. The role of hospital environment in transmissions of multidrug-resistant gram-negative organisms. *Antimicrob. Resist. Infect. Control* 9, 29 (2020).
- Chaoui, L., Mhand, R., Mellouki, F. & Rhallabi, N. Contamination of the surfaces of a health care environment by multidrug-resistant (MDR) bacteria. *Int. J. Microbiol.* 2019, 3236526 (2019).
- Dancer, S. J. Controlling hospital-acquired infection: focus on the role of the environment and new technologies for decontamination. *Clin. Microbiol. Rev.* 27, 665–690 (2014).
- 92. Hu, H. et al. Intensive care unit environmental surfaces are contaminated by multidrug-resistant bacteria in biofilms: combined results of conventional culture, pyrosequencing, scanning electron microscopy, and confocal laser microscopy. *J. Hosp. Infect.* **91**, 35–44 (2015).
- Russotto, V., Cortegiani, A., Raineri, S. M. & Giarratano, A. Bacterial contamination of inanimate surfaces and equipment in the intensive care unit. *J. Intensive Care* 3, 54 (2015).
- Boyce, J. M. Modern technologies for improving cleaning and disinfection of environmental surfaces in hospitals. *Antimicrob. Resist. Infect. Control* 5, 10 (2016).
- 95. CDC. Best practices for environmental cleaning in healthcare facilities in resource-limited settings (US Department of Health and Human Services, 2019).
- Cadnum, J. L. et al. Use of a stop valve to enhance disinfectant exposure may improve sink drain disinfection. *Infect. Control Hosp. Epidemiol.* 1–3, https:// doi.org/10.1017/ice.2018.318 (2018).
- Jones, J., Cadnum, J., Mana, T., Jencson, A. & Donskey, C. 1226. Application of a foam disinfectant enhances sink drain decontamination in hospital sinks. *Open Forum. Infect. Dis.*, 6, S440-S441 (2019).
- Kotsanas, D. et al. "Down the drain": carbapenem-resistant bacteria in intensive care unit patients and handwashing sinks. *Med. J. Aust.* 198, 267–269 (2013).
- Leitner, E. et al. Contaminated handwashing sinks as the source of a clonal outbreak of KPC-2-producing Klebsiella oxytoca on a hematology ward. *Antimicrob. Agents Chemother.* 59, 714–716 (2015).
- Soothill, J. S. Carbapenemase-bearing Klebsiella spp. in sink drains: investigation into the potential advantage of copper pipes. J. Hosp. Infect. 93, 152–154 (2016).
- 101. Kanamori, H. et al. Genomic analysis of multidrug-resistant Escherichia coli from North Carolina community hospitals: ongoing circulation of CTX-M-Producing ST131-H30Rx and ST131-H30R1 strains. Antimicrob. Agents Chemother. 61, https://doi.org/10.1128/AAC.00912-17 (2017).
- Pightling, A. W. et al. Interpreting whole-genome sequence analyses of foodborne bacteria for regulatory applications and outbreak investigations. *Front. Microbiol.* 9, 1482 (2018).
- 103. Lavenir, R. et al. Spatio-temporal analysis of infra-specific genetic variations among a Pseudomonas aeruginosa water network hospital population: invasion and selection of clonal complexes. J. Appl. Microbiol. 105, 1491–1501 (2008).
- 104. Snitkin, E. S. Contamination of hospital plumbing: a source or a sink for antibiotic-resistant organisms? JAMA Netw. Open 2, e187660 (2019).
- 105. Ferroni, A. et al. Outbreak of nosocomial urinary tract infections due to Pseudomonas aeruginosa in a paediatric surgical unit associated with tapwater contamination. *J. Hosp. Infect.* **39**, 301–307 (1998).

Acknowledgements

This work was supported in part by an award to J.H.K., G.D., and C.D.B. from the Agency for Healthcare Research and Quality (AHRQ) of the US Department of Health & Human Services (DHHS) (grant number R01HS027621); an award to G.D. by the National Institute of Allergy and Infectious Diseases (NIAID) of the National Institutes of Health (NIH) (grant number U01AI123394); and awards to J.H.K. from NIAID (grant number 1K23AI137321), Barnes-Jewish Hospital Foundation (award number 5102), and the Washington University Institute of Clinical and Translational Sciences grant UL1TR002345 from the National Center for Advancing Translational Sciences (NCATS) of the NIH (Award number 4462). K.V.S. was supported by a Research Scholar Award from the Society for Healthcare Epidemiology of America. The content is solely the responsibility of the authors and does not necessarily represent the official views of the funding agencies. The authors thank the nursing, patient care, and administrative staff of the SCT ICU at Barnes-Jewish Hospital and the Environmental Services staff at Barnes-Jewish Hospital, Lindsay Selner, MBA, MHA, BJC Healthcare Director of Activation Planning, Colleen McEvoy, MD, Director of the Stem Cell Transplant and Oncology Intensive Care Unit (SCT ICU) at Barnes-Jewish Hospital (BJH), Courtney McCullough, RN, BSN, Clinical Nurse Manager of the BJH SCT ICU, Mana Paolo Vinzon, RN, BSN, MBA, Assistant Clinical Nurse Manager of the BJH SCT ICU, all the nursing staff and environmental services staff at the BJH SCT ICU, William Buol, CEH, Senior Manager of Environmental Services at BJH, and Peter Westervelt, MD, PhD, Section Director of Bone Marrow Transplantation & Leukemia at Washington University. The authors thank the Edison Family Center for Genome Sciences & Systems Biology at WUSM staff, Eric Martin, Brian Koebbe, Jessica Hoisington-López, and MariaLynn Crosby for technical support in high-throughput sequencing and computing. The authors thank members of the Dantas lab for helpful general discussions and comments on the manuscript, and Tiffany Hink for her support of the Kwon Laboratory.

Author contributions

Conceptualization, C.D.B., G.D., J.H.K. Methodology, Investigation, Data Curation, and Analyses, K.V.S., E.P.N., C.C., M.A.W., C.J., J.F., S.S., M.H.B. Writing of Manuscript, K.V.S., E.P.N., C.D.B., G.D., J.H.K. Review and Editing of Manuscript, all authors.

Competing interests

The authors declare no competing interests.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-022-00124-5.

Correspondence and requests for materials should be addressed to Carey-Ann D. Burnham, Gautam Dantas or Jennie H. Kwon.

Peer review information *Communications Medicine* thanks the anonymous reviewers for their contribution to the peer review of this work.

Reprints and permission information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2022